Author's response to reviews

Title: Prediction of grip and key pinch strength in 978 healthy subjects.

Authors:

Felix Angst (fangst@vtxmail.ch)
Susann Drerup (sr.s@gmx.net)
Stephan Werle (stephan.werle@bluewin.ch)
Daniel D Herren (daniel.herren@kws.ch)
Beat R Simmen (beat.simmen@kws.ch)
Jörg Goldhahn (joerg.goldhahn@kws.ch)

Version: 3 Date: 8 April 2010

Author's response to reviews: see over
Prediction of grip and key pinch strength in 978 healthy subjects.
Angst F, Drerup S, Werle S, Herren DB, Simmen BR, Goldhahn J.

Comments to the revision 2

Reviewer 1: Richard Bohannon

1. Deletion of “after orthopedic interventions”
This has been deleted as requested in the 1st sentence of the Abstract, p. 2

2. “Few studies have examined…”
This has been corrected as requested the 3rd paragraph of the Background, p.3.

3. P.7, refer to Excel.
This has been removed as requested. The example of the exact programming in Excel has been left in the discussion, p.10 because this was requested by reviewer 2, Joy MacDermid in the first revision.